Author's response to reviews

Title:A cross-sectional analysis of the relationship between tobacco and alcohol outlet density and neighbourhood deprivation

Authors:

Niamh K Shortt (niamh.shortt@ed.ac.uk)
Catherine Tisch (catherine.tisch@ed.ac.uk)
Jamie Pearce (Jamie.pearce@ed.ac.uk)
Richard Mitchell (Richard.Mitchell@glasgow.ac.uk)
Elizabeth A Richardson (E.Richardson@ed.ac.uk)
Sarah Hill (S.E.Hill@ed.ac.uk)
Jeff Collin (jeff.collin@ed.ac.uk)

Version:2 Date:27 April 2015

Author's response to reviews:

Changes have been made. All authors emails are on the title page(s). Introduction is now title background, All tables are now after the references. A note has been added regarding ethical approval in the methods. It was also noted that the study did not require individual level consent as it analyses area level census data freely available. A list of abbreviations has been added.